Author’s response to reviews

Title: Comparison of Survival Benefits of Nephron-Sparing Intervention or Active Surveillance for Patients with Localized Renal Masses: A Systematic Review and Meta-Analysis

Authors:

Runqi GUO (lawlietkaku@gmail.com)
Xiao-Guang Li (xglee88@126.com)

Version: 2 Date: 19 Jul 2019

Author’s response to reviews:

1. Response to comment: Title page- Please provide email addresses of all co-authors on the title page.

Response: According to the comment, we have added the email addresses of all co-authors on the title page.

2. Response to comment: Section heading- Please rename the section heading 'Materials and Methods' to 'Methods' in the main text of this manuscript.

Response: The section heading "Materials and Methods" in the main text of the manuscript has been renamed to "Methods".

3. Response to comment: Clean copy- On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Response: All tracked changes and highlighting have been removed and the clean copy of the manuscript has been uploaded.